Author's response to reviews

Title: Brazilian Portuguese translation and cross-cultural adaptation of the "Caregiver Priorities and Child Health Index of Life with Disabilities" (CPCHILD) questionnaire

Authors:

Luiz A Pellegrino (luizpellegrino@gmail.com)
Erika VP Ortolan (epaiva@fmb.unesp.br)
Claudia S Magalhaes (claudi@fmb.unesp.br)
Ariane A Viana (arianeaviana@gmail.com)
Unni G Narayanan (unni.narayanan@sickkids.ca)

Version: 2
Date: 23 November 2013

Author's response to reviews: see over
Dear Editorial Board of BMC Pediatrics,

We are submitting the revised manuscript Brazilian Portuguese translation and cross-cultural adaptation of the “Caregiver Priorities and Child Health Index of Life with Disabilities” (CPCHILD©) questionnaire for consideration by the editors. All the co-authors have read and agreed to the changes. In this revised version, our purpose was to improve the manuscript based on the suggestions and comments provided by reviewers.

Description of the changes that were made:
- Corrections and proofreading to improve the quality of the written English.
- Figure and Table titles listed after the references.
- Inclusion of an additional file containing the changes generated at different points in the process of transcultural adaptation.
- Inclusion of a reference citing the PedsQL-CP module (a condition-specific HRQoL questionnaire for children and adolescents with CP).
- Restructuring of paragraph 1 of the Results section to improve comprehension of the points under discussion.
- Inclusion of the study that used the original questionnaire, reinforcing its contribution and relevance to the current understanding of the quality of life of children with cerebral palsy.

We believe that the comments provided by the reviewers have been addressed in this version and we are grateful to the reviewers for their contribution to improving the manuscript.

Sincerely yours,
Professor Erika Ortolan, on behalf of the authors.